Author's response to reviews

Title: A population-based study of ambulatory and surgical services provided by orthopaedic surgeons for musculoskeletal conditions

Authors:

Mayilee Canizares (mcanizar@uhnres.utoronto.ca)
Crystal MacKay (cmackay@uhnres.utoronto.ca)
Aileen M Davis (adavis@uhnres.utoronto.ca)
Nizar Mahomed (nmahomed@uhnres.utoronto.ca)
Elizabeth M Badley (e.badley@utoronto.ca)

Version: 7 Date: 30 March 2009

Author's response to reviews:

Minor changes were made to correct typos in the manuscript.